Author's response to reviews

Title: Aminoglycoside therapy for childhood urinary tract infection due to extended-spectrum beta-lactamase-producing Escherichia coli or Klebsiella pneumoniae

Authors:

Seung Beom Han (beomsid@catholic.ac.kr)
Sung Chul Lee (erdc87@naver.com)
Soo Young Lee (sylee@catholic.ac.kr)
Dae Chul Jeong (dcjeong@catholic.ac.kr)
Jin Han Kang (kjhan@catholic.ac.kr)

Version: 3 Date: 21 September 2015

Author's response to reviews: see over
Editor’s comment

On the basis of the comments we have received we feel that your manuscript is suitable for publication in the journal. However, before we proceed with acceptance there are some further editorial revisions which we require you to make. Please note that your manuscript cannot be accepted for publication until these revisions have been satisfactorily addressed.

----------------------------------------------------------------

1. Abbreviations

Please check that all abbreviations are defined when they are used for the first time in the text. Please also add a section 'List of abbreviations used' after section 'Conclusions'.

→ The new section “List of abbreviations used” was added in the manuscript (line ).

2. Acknowledgements

You may wish to consider adding a section ‘Acknowledgements’ before the reference list. There you can acknowledge anyone who made substantial contributions towards the article, but who does not meet the criteria for authorship, and the source(s) of funding for each author.

→ We have nothing to state in the “Acknowledgements.”

----------------------------------------------------------------

Authors sincerely thank the editor and reviewers for valuable advice.